MULTIDISCIPLINARY REHABILITATION FOR ADULTS WITH NEUROMYELITIS OPTICA SPECTRUM DISORDERS: A PILOT STUDY

Dong-Mei SUO, MD^{1,2}, Lin-lin LIU, MSc³, Kun JIA, MSc¹, Lin-jie ZHANG, MD¹, Li-Min LI, MD¹, Jing WANG, MD¹, Yuan QI, MSc¹, Hai-jie LIU, MD², Chun-Xiao WAN, MD² and Li YANG, MD¹

From the ¹Department of Neurology and Tianjin Neurological Institute, ²Department of Rehabilitation, Tianjin Medical University General Hospital, and ³Department of Paediatrics, Tianjin Hospital of ITCWM Nankai Hospital, Tianjin, China

Objective: To provide detailed data on the effects of multidisciplinary rehabilitation for patients with neuromyelitis optica spectrum disorder with moderate to severe disability.

Design: A pilot randomized control study.

Subjects/patients: A total of 39 patients with neuromyelitis optica spectrum disorder were randomized into intervention or control groups.

Methods: The intervention group received multidisciplinary rehabilitation 5 times/week for 4 weeks in a hospital, and the patients were guided to continue community- or home-based rehabilitation for 3 months. The control group did not receive any specific rehabilitation intervention. Disability was assessed using the Extended Disability Status Scale (EDSS) and Functional Systems (FS) scores after 4 weeks of rehabilitation and 3 months of follow-up.

Results: The mean EDSS score was 7.5 at admission for both groups. Improvements (p<0.05) in the EDSS score and domains of bowel, bladder and motor functions (pyramidal and walking function) were noted in the multidisciplinary rehabilitation group after 4 weeks. After 3 months, the patients in the usual care group showed improvement in EDSS score and walking ability score; however, no significant changes in other variables were noted.

Conclusion: These results suggest that multidisciplinary rehabilitation potentially promotes motor functional recovery in patients with neuromyelitis optica spectrum disorders.

Key words: neuromyelitis optica spectrum disorders; multidisciplinary rehabilitation; disability.

Accepted Jul 17, 2019; E-pub ahead of print Aug 16, 2019

J Rehabil Med 2019; 51: 692-697

Correspondence address: Li Yang, Department of Neurology and Tianjin Neurological Institute, Tianjin Medical University General Hospital, Tianjin 300052, China. E-mail: yangli2001@tmu.edu.cn

Neuromyelitis optica (NMO) is an inflammatory disease of the central nervous system that selectively targets the optic nerve and spinal cord (1). In 2007, the term NMO spectrum disorders (NMOSD) was introduced to include AQP4-IgG-seropositive patients with limited or inaugural forms of NMO (e.g. first-attack longitudinally extensive transverse myelitis (LETM) or recurrent or bilateral optic neuritis) who were at high risk of future attacks (2). NMOSD

LAY ABSTRACT

The aim of this study is to investigate the effects of multidisciplinary rehabilitation for patients with neuromyelitis optica spectrum disorders who have moderate to severe disability. In 39 patients with neuromyelitis optica spectrum disorders, disability was assessed after 4 weeks of rehabilitation and 3 months of follow-up. In the patients who underwent multidisciplinary rehabilitation, bowel and bladder and motor functions (pyramidal and walking function) were improved compared with those with usual care. These results confirm the shortterm effectiveness of multidisciplinary rehabilitation in patients with neuromyelitis optica spectrum disorders in terms of improving impairment. The results also suggest that multidisciplinary rehabilitation is a safe and feasible therapy for adults with neuromyelitis optica who have severe disability.

is relatively frequent worldwide. In Asia, NMOSD represents approximately 50% of all demyelinating disorders (3). NMOSD displays a relapsing-remitting course (2), and severe disability remained after immunotherapy with recurring relapse (4). Moreover, a large number of studies have shown that patients with multiple sclerosis, the most common demyelinating disorder of the central nervous system, can benefit from rehabilitation therapy (5–8). However, evidence regarding the benefits of rehabilitation in patients with NMOSD remains insufficient.

EDSS is the standard for evaluating the clinical treatment of multiple sclerosis, while NMO and multiple sclerosis (MS) are both demyelinating diseases of the central nervous system. However, few studies have used the Expanded Disability Status Scale (EDSS) to evaluate disability after multidisciplinary rehabilitation (MDR) in patients with NMOSD. In addition, the effect of MDR treatment on NMOSD is unclear. Hence, the aims of this study were: (*i*) to evaluate the effectiveness of MDR rehabilitation therapy in patients with NMOSD; and (*ii*) to explore which functional systems may benefit from MDR in patients with NMOSD.

MATERIALS AND METHODS

Participants and clinical parameters

A total of 39 consecutive patients with NMOSD were recruited from the Neurology Department of Tianjin Medical University General Hospital from January 2014 to November 2016. The diagnosis of NMOSD fulfilled the 2015 Wingerchuk criteria (9). The inclusion criteria for this study were: (*i*) disability as a result of NMOSD ($4 \le \text{EDSS}$ score ≤ 9); (*ii*) patient receiving 500 mg methylprednisolone intravenously, which was reduced by half every 3 days, followed by a gradual tapering until 40 mg was reached (14–16 days from the acute attack). The exclusion criteria were: (*i*) acute attack within 14 days; (*ii*) other conditions that may affect motor function or impaired cognitive functioning; (*iii*) severe concurrent diseases (i.e. cancer, cardiopulmonary diseases or severe psychiatric disorders). The subjects were divided randomly into 2 groups: a rehabilitation group (MDR) and a control group (usual care; UC). A random sequence was created by a computer. Patients in the UC refused or voluntarily ceased rehabilitation treatment (due to dissatisfaction or for economic reasons).

Demographic and clinical data, including sex, age, and disease duration, were collected. Informed consent was obtained from all participants, and the study was approved by the Tianjin Medical University General Hospital institutional review board and ethics committee.

Intervention

Multidisciplinary rehabilitation (MDR). Medical rehabilitation is defined as "a set of measures that assist individuals who experience disability to achieve and maintain optimal physical, sensory, intellectual, psychological and social functioning in interaction with their environment" (8). Since clinical manifestations of NMO and MS overlap and NMO was thought to be a variant of MS, some rehabilitation therapies in this study were co-opted from treatments used for MS and NMO.

For pyramidal function and walking ability, patients were offered physiotherapy. Physiotherapy typically involved at least 30 min of daily exercise. These exercises included active or passive range-of-motion exercises; strengthening exercises; hand function exercises; transfer exercises; balance training aimed at increasing muscle strength, improving balance and gait, decreasing spasticity and improving functional activities of daily living; and 20 min of robot-assisted gait training according to the patient's needs. In addition, occupational, cognitive, respiratory and/or phoniatric therapy were provided when necessary (10).

For sensory function, patients were provided sensory reeducation techniques and occupational therapy.

For bladder and bowel, patients received physical treatments, including pelvic floor exercises and/or intermittent cleaning and/ or self-catheterization (11, 12), according to their dysfunction.

Patients were allowed to rest before becoming fatigued. Multidisciplinary inpatient rehabilitation was performed 5 days/week for 2–4 h/day for 4 weeks. After 4-week inpatient rehabilitation, the patients were guided to continue community or home-based rehabilitation (walking training, wheelchair operation training) for 3 months.

Robot-assisted gait training: LokoHelp (Germany) (13) practical design offered therapists economically feasible access to quick and convenient body weight support (BWS), including ergonomically adjustable seating for assistance with manual treatment. This training was performed in the hospital.

Usual care. The control group (usual care; UC) did not receive any specific rehabilitation treatment for gait performance and mobility improvement.

During the entire study, both groups were authorized, at will, to exercise in non-rehabilitative contexts.

Outcome measures

NMOSD impairment was evaluated using EDSS and Functional Systems (FS) scores before the treatment and 4 weeks and 3 months after the treatment. EDSS is based on a detailed neurological examination that combines impairment and disability on a 10-step ordinal scale, ranging from 0 (normal) to 10 (death), and was performed by 2 neurologists who were both certified by Neurostatus for EDSS competency (14, 15). The patients' FS were described on 8 subscales, including visual (or optic) functions, brain stem functions, pyramidal functions, sensory functions, bowel and bladder functions, cerebellum functions, cerebral (or mental) functions, and walking functions. If the walking distance is <500 m, the EDSS score depends on walking ability (16). The EDSS scores were evaluated at the start and end of the training period.

Statistical analysis

The data for the MDR group were compared with the respective results of the parallel UC group. Descriptive statistics (median, minimum and maximum) are given for all variables. Non-parametric testing (Wilcoxon signed-rank) was performed to compare the EDSS and FS scores. A *p*-value of 0.05 or less was considered significant for all statistical analyses. All statistical analyses and graphs were performed using GraphPad PRISM 5 (Graph Pad Software Inc., San Diego, CA, USA).

RESULTS

Demographic and clinical features

A total of 39 patients with NMOSD were assessed for eligibility and randomized into 2 groups: an MDR group (n=21) and a UC group (n=18). One patient in the MDR group and one patient in the control group were lost at the 4-week assessment due to relapse. Three patients in the MDR group and 2 patients in the UC group dropped out due to refusal to continue, and these patients did not perform evaluations at 3 months (Fig. 1). The patients' demographic and clinical characteristics are reported in Table I. In total, 39 patients with NMOSD (29 females, 10 males) were included in the study. A final total of 32 patients completed the study. The median EDSS score at admission was 7.5 (range 6.0–9.0) All patients with NMOSD had lesions in the spinal cord. No statistically significant baseline

Table I. Baseline characteristics of people with neuromyelitis optica

 spectrum disorders allocated to multidisciplinary rehabilitation

 (MDR) or usual care

Patients' characteristics	MDR n = 21	Usual care $n = 18$
Age, year, mean (SD) Number of women, <i>n</i> (%) Number of relapses Duration months, mean (SD)	51.5 (14.0) 15 (88) 4.7 28 (22.9)	54.5 (7.8) 14 (93) 5.2 26 (32.5)

SD: standard deviation.



Fig. 1. CONSORT diagram for multimodal exercise training intervention.

differences between 2 groups were found at baseline (Fig. 2).

Outcomes after rehabilitation

All patients completed high-dose intravenous methylprednisolone therapy (2–3 weeks from the acute attack) and underwent an oral steroid tapering period to achieve 10–15 mg/day.

Table II shows the median EDSS and FS scores for the MDR and UC groups before treatment. No sta-

Table II. Extended Disability Status Scale (EDSS) and Functional

 System scores to multidisciplinary rehabilitation (MDR) or usual

 care (UC) before treatment

EDSS and EDSS subscales	MDR <i>n</i> = 21 Median (range)	Usual care $n = 18$ Median (range)
EDSS	7.5 (6.0-8.5)	7.5 (6.0-8.0)
Walking ability	7.5 (6-8.5)	7.5 (6.0-8.0)
Pyramidal function	4 (1-4)	4 (2-4)
Sensory function	3 (1-4)	3 (1-4)
Bowel and bladder	2 (1-4)	2 (1-4)
Cerebral function	2 (1-2)	2 (1-2)



Fig. 2. Extended Disability Status Scale (EDSS) and Functional System (FS) scores for the multidisciplinary rehabilitation (MDR) (n=21) or usual care (UC) (n=18) groups before and after 4 or 12 weeks of MDR treatment. Numbers are statistically reported as the mean and standard deviation (SD). ***p < 0.001; **p < 0.01; **p < 0.05. FS-WA: walking function. FS-PF: pyramidal functions. FS-SF: sensory functions. FS-BE bowel and bladder functions. FS-CF: cerebral (or mental) functions.

Table III. Changes in Extended Disability Status Scale (EDSS) and	d Functional System (FS) scores after treatment
---	---

	Pre-MDR	4 weeks after MDR	3 months after MDR ^c			
EDSS subscales	5 subscales Median (range) Median (range)		Median (range)	p-value ^a	p-value ^b	p-value ^c
MDR group ($n = 17$)						
EDSS scores	7.5 (6-8)	6.5 (5-7.5)	6 (2-7.5)	< 0.001	< 0.001	0.083
Walking ability	7.5 (6-8.5)	7.0 (5-8)	7.0 (2-8)	0.003	0.002	0.066
Pyramidal function	4 (1-4)	3 (1-4)	3 (1-4)	< 0.001	< 0.001	0.002
Sensory function	3 (1-4)	3 (1-4)	3 (1-4)	0.083	0.083	0.317
Bowel and bladder	2 (1-4)	1.5 (0-3)	1 (0-2)	0.001	< 0.001	0.059
Cerebral function	2 (1-2)	1 (1-2)	1 (0-2)	0.083	0.002	0.002
UC group $(n=15)$						
EDSS total scores	7.5 (6-8.5)	7.5 (6-8)	7.5 (6-8)	0.083	0.023	0.499
Walking ability	7.5 (6-8.5)	7.5 (6-8)	7.5 (6-8)	0.083	0.023	0.157
Pyramidal function	4 (2-4)	3 (2-4)	3 (2-4)	0.083	0.014	0.083
Sensory function	3 (1-4)	3 (1-4)	3 (1-4)	0.083	0.083	0.317
Bowel and bladder	2 (1-4)	2 (1-4)	2 (1-4)	0.059	0.059	0.317
Cerebral function	2 (1-2)	1 (1-2)	1 (1-2)	0.317	0.157	0.317

^aComparison between Pre-MDR and 4 weeks after MDR. ^bComparison between Pre-MDR and 3 months after MDR. ^cComparison between 4 weeks after MDR and 3 months after MDR. MDR: multidisciplinary rehabilitation; UC: usual care. Significant values are shown in bold.



Fig. 3. Extended Disability Status Scale (EDSS) and Functional System (FS) scores for multidisciplinary rehabilitation (MDR) (n = 21) before and after 4 or 12 weeks of MDR treatment. Numbers are statistically reported as mean and standard deviation (SD). ***p < 0.001; **p < 0.01; **p < 0.05. FS-WA: walking function. FS-PF: pyramidal functions. FS-SF: sensory functions. FS-BB: bowel and bladder functions. FS-CF: cerebral (or mental) functions.

tistically significant baseline differences were found between the 2 groups (p > 0.05). EDSS and FS scores are shown in Table III and Fig. 3.

The results showed significant differences between the 2 groups in EDSS total scores, walking ability scores. Furthermore, the patients in the MDR group recovered faster than those in the UC group. Neither the patients in the MDR group nor those in the UC group showed a statistically significant change in median scores for sensory function.

Following the 4-week MDR treatment, a significant improvement in the EDSS was observed compared with that at admission in the MDR group (p < 0.001). Moreover, in the subscale analyses, significant improvements were demonstrated in bowel and bladder function, pyramidal function, and walking function scores in patients in the MDR group (2 to 1.5, p=0.001; 4.0 to 3.0, p < 0.001; and 7.5 to 7.0, p=0.003; respectively), while sensory function and cerebral (or mental) function remained unchanged (3 to 3, p=0.083; 2.0 to 1.0, p=0.083; respectively). Furthermore, after 3 months of treatment, the trends in the FS were the same as those at discharge, except that fatigue in the cerebral (or mental) function score had improved (from 2.0 to 1.0, p=0.002; Table IV).

In the UC group, no significant differences in the EDSS scores and FS scores were noted after 4 weeks (p > 0.05). However, significant improvements in the EDSS scores, pyramidal function, and walking function scores were noted after 3 months.

Table shows the differences in the changes in median scores between the 2 groups at 4 weeks, 3 months and baseline.

DISCUSSION

This study demonstrated the benefit of MDR in patients with disability resulting from NMOSD, as assessed by EDSS and FS. The study also suggested that MDR was effective in improving the clinical status of patients with NMOSD in terms of impairments. In the UC group, the EDSS score did not change at 4 weeks and improved at 3 months,

which indicated that patients with MNOSD are taking corticosteroids (1 month) and exhibit a certain degree of self-healing ability (3 months) (17). The EDSS score improved at 4 weeks after treatment in the MDR group, which indicated that rehabilitation treatment promotes or speeds up the patient's recovery process. Moreover, the results indicated that the bowel and bladder, pyramidal, and walking functions of patients with NMOSD might recover after 4 weeks with MDR treatment.

To the best of our knowledge, this is the first study investigating which FS may benefit more from MDR treatment in an NMO population in MDR programmes. Three previous studies (2 case reports and 1 research article) described the rehabilitation outcomes of patients with NMO. One case report on the effect of vibration stimulation (VS) on NMO discovered that standing up and walking performance improved with VS, and this finding may support the hypothesis of neuronal plasticity on central pattern generators by VS (18). Plasticity is a result of the creation of new circuitry

Table IV. Extended Disability Status Scale (EDSS) and Functional System (FS) scores to multidisciplinary rehabilitation (MDR) or usual care (UC) 4 week after MDR

EDSS and EDSS subscales	Between-group differences								
	0-4 weeks ∆score Median (range)			0−3 months ∆score Median (range)			4 weeks-3 months ∆score Median (range)		
	UC	MDR	<i>p</i> -value	UC	MDR	<i>p</i> -value	UC	MDR	<i>p</i> -value
EDSS	0 (0-0.5)	0.5 (0-3)	< 0.001	0 (0-0.5)	1 (0-6)	< 0.001	0 (0-0.5)	0 (0-3)	0.583
Walking ability	0 (0-0.5)	0.5 (0-2.5)	0.014	0 (0-0.5)	0.5 (0-5.5)	0.023	0 (0-0.5)	0 (0-3)	0.576
Pyramidal function	0 (0-1)	1 (0-2)	0.002	0 (0-1)	1 (0-3)	0.001	0 (0-1)	1 (0-1)	0.028
Sensory function	0 (0-1)	0 (0-1)	0.876	0 (0-1)	0 (0-1)	0.812	0 (0-0)	0 (0-1)	0.348
Bowel and bladder	0 (0-1)	1 (0-2)	0.450	0 (0-2)	1 (0-2)	0.683	0 (0-1)	0 (0-2)	0.005
Cerebral function	0 (0-1)	0 (0-1)	0.356	0 (0-1)	1 (0-2)	0.008	0 (0-1)	0 (0-1)	0.002

through sprouting and alterations of synaptic strength. Appropriate rehabilitation could modify and enhance the plasticity process (19). Vahdat and colleagues provided strong evidence for local spinal plasticity over the course of human motor learning using functional magnetic resonance imaging (20). Similarly, Nechemia et al. (21) described 15 in-patients with NMO and 32 in-patients with MS who received MDR and found that both groups benefitted. Moreover, at discharge, the NMOSD group showed greater improvement in FIM scores and lower EDSS scores.

The current study investigated bowel and bladder, cerebral (or mental), pyramidal, and walking functions, which may benefit more from MDR. However, sensory dysfunction required the longest period for recovery. Intermittent catheterization, suprapubic tapping, and physical treatments, including pelvic floor exercises, have been applied to aid voiding for urinary retention. Thus, bowel and bladder functions were improved. Similarly, a systematic review reported that physical therapy techniques were effective for urinary disorders in MS populations with mild disability (22).

Walking function is a central outcome of rehabilitation intervention (23). Grasso and colleagues suggested that rehabilitation for patients with MS should be initiated early while the patient can walk independently or with assistance and lacks severe sphincteric and cognitive deficits (23). Accordingly, in this study, patients with NMOSD received MDR immediately after they completed high-dose intravenous methylprednisolone therapy (2–3 weeks after the acute attack). The regimen achieved a good effect, even compared with previous trials implementing rehabilitation in the MS population (24). In the current study, patients' disability levels were more severe (EDSS score ≥ 6). Furthermore, there were no adverse events during the entire rehabilitation programme, indicating that it is relatively safe and effective to introduce MDR early to adults with NMOSD despite severe disability. Nechemia et al. reported that inpatient MDR programmes available for patients with MS might be implemented successfully for patients with NMO (21). In the control group, some patients exhibited improvements in walking ability scores and pyramidal function. We hypothesize that this therapy may affect the natural course of the disease.

Sensory function recovery may occur over a long time, and these changes may have cumulative effects. Given the poor results for sensory function rehabilitation, we hypothesize that 4 weeks is not sufficient for the amelioration of this disability, but continued therapy may offer promising results.

None of the patients in our study had cognitive impairment (MMSE score > 27), but all had fatigue to different degrees. After 4 weeks of MDR, no obvious

changes in patient fatigue were observed. This finding suggested that rehabilitation training did not exacerbate fatigue. However, after 3 months, the fatigue was resolved to varying degrees. Rehabilitation with a focus on relieving fatigue via an improvement in mood and cardiovascular reconditioning was previously reported by Judica et al. and Romberg et al. (25, 26). Therefore, our results indicated that training for patients with NMOSD should be continued in both hospital and home or community settings in order to maintain these benefits.

In this study, EDSS was applied to the evaluation of NMOSD. The EDSS provided a detailed evaluation (the 8 major system dysfunctions in MS are involved) (27). However, the EDSS has the disadvantage that it depends only on walking function in severe dysfunction (e.g. EDSS ≥ 6 points); therefore, the EDSS score cannot reflect the situation in which the walking function has not changed but other system functions have improved. Therefore, there is a need to modify the EDSS for use in the evaluation of NMOSD.

Study limitations

This study has some limitations. Most of the patients exhibited more serious disability (EDSS \geq 6), which may affect the overall rehabilitation outcome (see above). Furthermore, NMOSD is a rare disease; therefore, the sample was small. A well-designed study should be performed on MDR of patients with NMOSD at different levels of disability. In future clinical trials, studies should compare the effectiveness of MDR in treating different degrees of NMOSD, and further research is required into the rationality of EDSS as an indicator for evaluating NMOSD.

Conclusion

This study confirmed the short-term effectiveness of MDR in patients with NMOSD in terms of improving impairment, as determined using the EDSS. Bowel and bladder, cerebral (or mental), pyramidal, and walking functions may benefit more from MDR, while sensory dysfunction required the longest period for recovery. The control group showed some improvement in walking ability scores and pyramidal function, although this improvement may be an effect of the natural course of disease.

Although this study is a pilot trial, the results suggest that MDR is a safe and feasible therapy for adults with NMOSD with severe disability. These primary observations will be helpful to all medical practitioners treating MDR in patients with NMOSD, and they advocate that patients with NMOSD should be candidates for early MDR in order to prevent the progression of neurological disability. A randomized

controlled clinical trial should be performed to confirm these preliminary findings.

ACKNOWLEDGEMENTS

The authors wish to thank all participants in this research and all the staff and students involved in data collection.

This study was supported by the Tianjin Research Program of Application Foundation and Advanced Technology (15JCZDJC35700), the National Science Foundation of China (81171363 to L.Y.), and the National Key Clinical Specialty Construction Project of China.

REFERENCES

- 1. Lennon VA, Wingerchuk DM, Kryzer TJ, Pittock SJ, Lucchinetti CF, Fujihara K, et al. A serum autoantibody marker of neuromyelitis optica: distinction from multiple sclerosis. Lancet 2004; 364: 2106-2112.
- 2. Wingerchuk DM, Lennon VA, Lucchinetti CF, Pittock SJ, Weinshenker BG. The spectrum of neuromyelitis optica. Lancet Neurol 2007; 6: 805-815.
- 3. Kowarik MC, Soltys J, Bennett JL. The treatment of neuromyelitis optica. J Neuroophthalmol 2014; 34: 70-82.
- 4. Kikuchi S, Fukazawa T. "OSMS is NMO, but not MS": confirmed by NMO-IgG? The Lancet Neurology 2005; 4: 594–595.
- 5. Latimer-Cheung AE, Martin Ginis KA, Hicks AL, Motl RW, Pilutti LA, Duggan M, et al. Development of evidence-informed physical activity guidelines for adults with multiple sclerosis. Arch Phys Med Rehabil 2013; 94: 1829-1836.e7.
- 6. Sá MJ. Exercise therapy and multiple sclerosis: a systematic review. J Neurol 2014; 261: 1651-1661.
- 7. Salhofer-Polanyi S, Windt J, Sumper H, Grill H, Essmeister M, Diermayr G, et al. Benefits of inpatient multidisciplinary rehabilitation in multiple sclerosis. NeuroRehabilitation 2013; 33: 285-292.
- 8. Khan F, Amatya B. Rehabilitation in multiple sclerosis: a systematic review of systematic reviews. Arch Phys Med Rehabil 2017; 98: 353-367.
- 9. Wingerchuk DM, Banwell B, Bennett JL, Cabre P, Carroll W, Chitnis T, et al. International consensus diagnostic criteria for neuromyelitis optica spectrum disorders. Neurology 2015; 85: 177-189.
- 10. Pompa A, Morone G, Iosa M, Pace L, Catani S, Casillo P, et al. Does robot-assisted gait training improve ambulation in highly disabled multiple sclerosis people? A pilot randomized control trial. Mult Scler 2017; 23: 696-703.
- 11. Marola S, Ferrarese A, Gibin E, Capobianco M, Bertolotto A, Enrico S, et al. Anal sphincter dysfunction in multiple sclerosis: an observation manometric study. Open Med (Wars) 2016; 11: 509-517.
- 12. Castel-Lacanal E, Gamé X, Clanet M, De Boissezon X, Brassat D, Rischmann P, Marque P. Assessment of a program to encourage the multidisciplinary management of

urinary disorders in multiple sclerosis. Neurourol Urodyn 2017: 36: 706-709.

- 13. Freivogel S, Mehrholz J, Husak-Sotomayor T, Schmalohr D. Gait training with the newly developed 'LokoHelp'-system is feasible for non-ambulatory patients after stroke, spinal cord and brain injury. A feasibility study. Brain Inj 2008; 22: 625-632.
- 14. D'Souza M, Yaldizli Ö, John R, Vogt DR, Papadopoulou A, Lucassen E, Menegola M, Andelova M, Dahlke F, Schnyder F, Kappos L. Neurostatus e-Scoring improves consistency of Expanded Disability Status Scale assessments: A proof of concept study.Mult Scler 2017; 23: 597-603.
- 15. Şen S.Noro PsikiyatrNeurostatus and EDSS Calculation with Cases. Ars 2018; 55: S80-S83.
- 16. Rabadi MH, Vincent AS, Comparison of the Kurtkze expanded disability status scale and the functional independence measure: measures of multiple sclerosis-related disability. Disability and rehabilitation 2013; 35: 1877-1884.
- 17. Nedeljkovic U, Raspopovic ED, Ilic N, Vujadinovic ST, Soldatovic I, Drulovic J. Effectiveness of rehabilitation in multiple sclerosis relapse on fatigue, self-efficacy and physical activity. Acta Neurol Belg 2016; 116: 309-315.
- 18. Lin HN, Nagaoka M, Hayashi Y, Hatori K. Effect of vibration stimulation on dysbasia of spastic paraplegia in neuromyelitis optica: a possible example of neuronal plasticity. BMJ Case Rep 2012; pii: bcr2012006793.
- Rasova K, Prochazkova M, Tintera J, Ibrahim I, Zimova D, 19. Stetkarova I, et al. Motor programme activating therapy influences adaptive brain functions in multiple sclerosis: clinical and MRI study. Int J Rehabil Res. 2015; 38: 49-54.
- 20. Vahdat S, Lungu O, Cohen-Adad J, Marchand-Pauvert V, Benali H, Dovon J, Simultaneous brain-cervical cord fMRI reveals intrinsic spinal cord plasticity during motor sequence learning. PLoS Biol 2015; 13: e1002186.
- 21. Nechemia Y, Moreh E, Weingarden H, Bloch A, Givon U, Vaknin-Dembinsky A, et al. Effectiveness of multi-disciplinary rehabilitation for patients with neuromyelitis optica. J Spinal Cord Med 2016; 39: 311-316.
- 22. Gaspard L, Tombal B, Castille Y, Opsomer RJ, Detrembleur C. Troubles du bas appareil urinaire chez des patients atteints de la sclérose en plaques et kinésithérapie pelvi-périnéale: revue systématique. Prog Urol 2014; 24: 222-228.
- 23. Learmonth YC, Dlugonski DD, Pilutti LA, Sandroff BM, Motl RW. The reliability, precision and clinically meaningful change of walking assessments in multiple sclerosis. Mult Scler 2013; 19: 1784-1791.
- 24. Wonneberger M, Schmidt S. Changes of gait parameters following long-term aerobic endurance exercise in mildly disabled multiple sclerosis patients: an exploratory study. Eur J Phys Rehabil Med 2015; 51: 755-762.
- 25. Judica E, Martinelli Boneschi F, Ungaro D, Comola M, Gatti R, Comi G. Rossi P. Impact of fatigue on the efficacy of rehabilitation in multiple sclerosis. J Neurol 2011; 258: 835-839.
- Romberg A, Ruutiainen J, Puukka P, Poikkeus L. Fatigue in 26 multiple sclerosis patients during inpatient rehabilitation. Disabil Rehabil 2008; 30: 1480-1485
- 27. Kurtzke JF. On the origin of EDSS. Mult Scler Relat Disord 2015; 4: 95-103.